Author's response to reviews

Title: Improved Healing Response in Delayed Unions of the Tibia with Low-Intensity Pulsed Ultrasound: Results of a Randomized Sham-Controlled Trial

Authors:

Markus D Schofer M.D. (schofer@med.uni-marburg.de)
Jon E Block Ph.D. (proxy@jonblockphd.com)
Julia Aigner M.D. (julia.aigner@klinik-heidenheim.de)
Andreas Schmelz M.D. (andreas.schmelz@klinik-heidenheim.de)

Version: 3 Date: 31 August 2010

Author's response to reviews: see over
30 August 2010

The BMC Series Editorial Production Team
BMC Musculoskeletal Disorders

RE: MS 9464881183921728

Dear Production Team:

My coauthors and I are delighted you have accepted our paper entitled “Improved healing response in delayed unions of the tibia with low-intensity pulsed ultrasound: results of a randomized sham-controlled trial.”

We also have received confirmation that the associated production charges have been paid in full.

At your request, we have made the following formatting changes to the manuscript to conform to journal style:

1) The Table and Figure have been edited to include an Arabic numeral and a more distinct title for the Table. The mention of the Table in the text has also been altered accordingly.

2) The title has been changed to lower case.

3) We have modified the text throughout to conform with BMC style including the reference list.

4) We have removed qualifications from the author list.

This manuscript should now be in fine shape for publication.

Best Regards,

Jon E. Block, Ph.D.
Corresponding author